Author’s response to reviews

Title: Myocardial Inflammation, Injury and Infarction During On-Pump Coronary Artery Bypass Graft Surgery

Authors:

Shirjel Alam (s.r.alam@sms.ed.ac.uk)
Colin Stirrat (cstirrat@exseed.ed.ac.uk)
Nicholas Spath (nspath@nhs.net)
Vipin Zamvar (zamvarv@hotmail.com)
Renzo Pessotto (Renzo.Pessotto@nhslothian.scot.nhs.uk)
Marc Dweck (Marc.Dweck@nhslothian.scot.nhs.uk)
Colin Moore (Colin.Moore@luht.scot.nhs.uk)
Scott Semple (scott.seemple@ed.ac.uk)
Nicholas Mills (Nick.Mills@ed.ac.uk)
Anoop Shah (Anoop.Shah@ed.ac.uk)
Saeed Mirsadraee (s.mirsadraee@imperial.ac.uk)
David Newby (D.E.Newby@ed.ac.uk)
Peter Henriksen (phenrik1@staffmail.ed.ac.uk)

Version: 2 Date: 19 Nov 2017

Author’s response to reviews:

Dear sir/madam.

Many thanks for reviewing this paper. Although the author's portal states a minor correction is required, the actual attached letter states the paper has been accepted for publication. All revisions previously requested have been performed.
I have therefore simply corrected the authors list and re-submitted the paper. I hope this is acceptable